Author's response to reviews

Title: Association of total daily physical activity with disability in community-dwelling older persons: a prospective cohort study

Authors:

Raj C Shah (Raj_C_Shah@rush.edu)
Aron S Buchman (Aron_S_Buchman@rush.edu)
Patricia A Boyle (Patricia_A_Boyle@rush.edu)
Sue Leurgans (Sue_E_Leurgans@rush.edu)
David A Bennett (David_A_Bennett@rush.edu)

Version: 2 Date: 25 May 2012

Author's response to reviews:

Thank you for the opportunity to adjust the abstract and manuscript format. We carefully reviewed the online instructions for authors regarding the preparation of a research manuscript. We have added a "Methods" section in the Abstract. We have moved the "Methods" section in the manuscript so that it occurs after the "Background" section and before the "Results" section. We have updated the ordering of our citations to match the new formatting changes.